Author's response to reviews

Title: Male gonadal dose of ionizing radiation delivered during X-ray examinations and monthly probability of pregnancy: a population-based retrospective study.

Authors:

Sandra Sinno-Tellier (sinno@vjf.inserm.fr)
Jean Bouyer (bouyer@vjf.inserm.fr)
Beatrice Ducot (ducot@vjf.inserm.fr)
Beatrice Geoffroy-Perez (b.geoffroyperez@invs.sante.fr)
Alfred Spira (spira@vjf.inserm.fr)
Remy Slama (slama@vjf.inserm.fr)

Version: 4 Date: 8 February 2006

Author's response to reviews:

Dear Editor,

Thank you for your comments.

We revised our manuscript in accordance with the formatting checklist and added a sentence about the consent of the study population in the paragraph of methods (p5 l19).

Yours sincerely,

Dr Sinno-Tellier